**Author's response to reviews**

**Title:** Interest in neurology during medical clerkship in three Nigerian medical schools

**Authors:**

Emmanuel O Sanya (emanuelosanya@yahoo.com)
Olugbenga E Ayodele (gbengaayo@yahoo.ca)
Timothy O Olanrewaju (timothysegun@yahoo.com)

**Version:** 6  **Date:** 11 May 2010

**Author's response to reviews:**

Dear Editor

BMC Medical Education,

In line with the latest comments sent to us we have done the necessary corrections suggested and these are;

1. the name of the 3rd author on the submitted system and article now the same corrected
2. the abstract now corrected to in-house format- background, methods, results and discussions.

we thanks the editorial team for their kind review of our article.

Thanks

Dr EO Sanya